



## BAISHIDENG PUBLISHING GROUP INC

8226 Regency Drive, Pleasanton, CA 94588, USA

Telephone: +1-925-223-8242

Fax: +1-925-223-8243

E-mail: [bpgoffice@wjgnet.com](mailto:bpgoffice@wjgnet.com)

<http://www.wjgnet.com>

---

### ESPS PEER-REVIEW REPORT

**Name of journal:** World Journal of Clinical Pediatrics

**ESPS manuscript NO:** 24468

**Title:** Prevalence of recent immunisation in children with febrile convulsions

#### COMMENTS TO AUTHORS

Good study provided record based immunization history is of quality. Confidence interval are very wide and table of results of logistic regression may be added.

Thank you for your comments, however, given the small number of results we do not think it is of value to repeat the results in a table.



## BAISHIDENG PUBLISHING GROUP INC

8226 Regency Drive, Pleasanton, CA 94588, USA

Telephone: +1-925-223-8242

Fax: +1-925-223-8243

E-mail: [bpgoffice@wjgnet.com](mailto:bpgoffice@wjgnet.com)

<http://www.wjgnet.com>

---

### ESPS PEER-REVIEW REPORT

**Name of journal:** World Journal of Clinical Pediatrics

**ESPS manuscript NO:** 24468

**Title:** Prevalence of recent immunisation in children with febrile convulsions

#### COMMENTS TO AUTHORS

Interesting idea and useful information if positive findings. Significantly hampered by lack of recording regarding vaccination in notes. Would have been better if other record correlated eg state or national vaccination database.

Thank you for your comments. We agree that this would have been better if based on a state or national database, we have made a comment in the Discussion that this should be a future plan using linked data sets from state databases of immunization history and emergency department presentations.